Author's response to reviews

Title: Self-assessed health among Thai elderly

Authors:

Fariha Haseen (far_haseen@yahoo.com)
Ramesh Adhikari (rameships@gmail.com)
Kusol Soonthorndhada (prkst@mahidol.ac.th)

Version: 4 Date: 23 May 2010

Author's response to reviews: see over
Dear Editorial Production Team

Thank you very much for your valuable comments. We have incorporated all the comments in our revised manuscript. Please find point by point responses of the comments.

Regards,
Fariha Haseen

Major revisions (we require the author to make these changes)

Comments: Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Response: We have added “all authors read and approved the final manuscript” in our revised manuscript.

Comments: References: Please make sure references are cited in numerical order. We notice reference 12 is cited after reference 13.

Response: We have made correction.

Comments: References: reference 39, is missing from the reference list please provide details whilst ensuring references are still cited in consecutive order. There is no reference 38 in the list or cited in the text so renumber this reference 38.

Response: We have made correction.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Comments: Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Response: We have removed all highlighting and tracking from the revised manuscript.

Comments: Tables- please remove the visible vertical lines from your tables

Response: We have removed the visible vertical lines from tables.

Comments: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: We have found some typography error and made correction in our revised manuscript.

Once again, thank you so much for giving opportunity to correct typo-error.